\documentclass[10pt]{article}

\usepackage{fullpage}
\usepackage{setspace}
\usepackage{parskip}
\usepackage{titlesec}
\usepackage[section]{placeins}
\usepackage{xcolor}
\usepackage{breakcites}
\usepackage{lineno}
\usepackage{hyphenat}





\PassOptionsToPackage{hyphens}{url}
\usepackage[colorlinks = true,
            linkcolor = blue,
            urlcolor  = blue,
            citecolor = blue,
            anchorcolor = blue]{hyperref}
\usepackage{etoolbox}
\makeatletter
\patchcmd\@combinedblfloats{\box\@outputbox}{\unvbox\@outputbox}{}{%
  \errmessage{\noexpand\@combinedblfloats could not be patched}%
}%
\makeatother


\usepackage[round]{natbib}
\let\cite\citep




\renewenvironment{abstract}
  {{\bfseries\noindent{\abstractname}\par\nobreak}\footnotesize}
  {\bigskip}

\titlespacing{\section}{0pt}{*3}{*1}
\titlespacing{\subsection}{0pt}{*2}{*0.5}
\titlespacing{\subsubsection}{0pt}{*1.5}{0pt}


\usepackage{authblk}


\usepackage{graphicx}
\usepackage[space]{grffile}
\usepackage{latexsym}
\usepackage{textcomp}
\usepackage{longtable}
\usepackage{tabulary}
\usepackage{booktabs,array,multirow}
\usepackage{amsfonts,amsmath,amssymb}
\providecommand\citet{\cite}
\providecommand\citep{\cite}
\providecommand\citealt{\cite}
% You can conditionalize code for latexml or normal latex using this.
\newif\iflatexml\latexmlfalse
\providecommand{\tightlist}{\setlength{\itemsep}{0pt}\setlength{\parskip}{0pt}}%

\AtBeginDocument{\DeclareGraphicsExtensions{.pdf,.PDF,.eps,.EPS,.png,.PNG,.tif,.TIF,.jpg,.JPG,.jpeg,.JPEG}}

\usepackage[utf8]{inputenc}
\usepackage[english]{babel}










\begin{document}

\title{Forensic Document Examination}



\author[1]{Timothy Campbell}%
\affil[1]{Up-Goer Five}%


\vspace{-1em}



  
  \date{May 26, 2024}


\begingroup
\let\center\flushleft
\let\endcenter\endflushleft
\maketitle
\endgroup








\sloppy


\selectlanguage{english}
\begin{figure}[h!]
\begin{center}
\includegraphics[width=0.70\columnwidth]{figures/ink-on-paper---50x---3D/ink-on-paper---50x---3D}
\caption{{A really, really close and deep picture of (aka Laser-scanning confocal
3D micrograph of inkjet ink (right) on paper fibres (left))
{\label{955097}}%
}}
\end{center}
\end{figure}

{Sometimes people do bad things and make papers that are not true to
make other people believe wrong stuff. These bad papers can be used to
take money from other people or a money house, say mean words to other
people, or make people think that the person is really someone else.
These bad papers are looked at by people who took years to learn all the
different ways to make bad papers, and know the signs of what bad things
to look for, even very tiny hidden things, that can only be seen with
colored lights and a big eye thing full of glass, that most people don't
know about. The person looks at the paper, the wet stuff that came from
a stick and means things, and the picture or words that a number-box
told a desk-box to make with more wet (or dry) stuff. A group of papers
are then written and put together for an old person in a black dress
(who used to talk a lot) to look at in a big room from a high chair.
Sometimes the person who looks for the bad stuff on papers goes and
talks to the old person, who might have 12 friends sitting in the big
room too. If the papers are bad, the old person hits a wood thing on a
desk to send the bad person to live in the house with high walls and bad
food. But sometimes the bad papers are not bad after all, so the person
is actually good and gets to stay out of the house with high walls and
instead lives in their own house, maybe eating good food, where they are
more happy. Back at the person's work desk, they will try to learn new
ways to do better work and make more people happy and safe from bad
papers.}

\par\null

\selectlanguage{english}
\FloatBarrier
\end{document}

